Author's response to reviews

Title: systemic lupus erythematosus presented with eosinophilic enteritis: a case report

Authors:

  mehrnаз asadi (asadi_m@tums.ac.ir)
  pejхman abdollahi (abdollahip@student.tums.ac.ir)
  mohhammad kalani (mohamadkalani@yahoo.com)
  masoud sotoudeh (masoud.sotoudeh.md@gmail.com)

Version: 7 Date: 12 January 2011

Author's response to reviews: see over
A cover letter considering the reviewer’s report

Title: systemic lupus erythematosus presented with eosinophilic enteritis: a case report

Version: 6 Date: 3 November 2010

Comments to authors:
The authors have added further materials to their case report that strengthens the presentation. There is improvement in the description of SLE criteria met by the patient. From the new information, it is interesting that the patient had a negative ANA (found in only about 1% of lupus cases). Therefore, the presentation of this patient is highly unusual.
Overall, the case report is of considerable interest and the documentation is improved. Unfortunately, some items that could strengthen the report like a picture of the small bowel biopsy, the discoid lupus skin lesion, or the renal biopsy will apparently just not be available.
The problem of multiple grammatical errors and unidiomatic English continues and requires substantial editing.

Quality of written English: Needs some language corrections before being published

Declaration of competing interests: 'I declare that I have no competing interests.

• Authors’ response

Some language and grammatical corrections were carried out. I asked help from a native speaker to make corrections. All corrections are in blue color.

The section of abdominal Ct scan and photomicrographs of colonic mucosa are included. I apologize for undue delay in providing the pictures.

My three colleagues and I declare that we have no competing interest.

At last, I really thank you for your valuable comments.

King regards

Mehrnaz Asadi